Author’s response to reviews

Title: Effects of media-assisted therapeutic approaches on physical activity of obese adults: A systematic review

Authors:

Alexandra Ziegeldorf (alexandra.ziegeldorf@uni-leipzig.de)

Petra Wagner (petra.wagner@uni-leipzig.de)

Hagen Wulff (hagen.wulff@uni-leipzig.de)

Version: 2 Date: 27 Jan 2020

Author’s response to reviews:

The authors would like to thank the Editor and the reviewers for their invaluable assistance and helpful comments. We noticed all annotations and revised the text accordingly. The reviewers support helps to improve the quality significantly.

Editor Comments

Comment 1:

“According to the reviewer comment, the manuscript needs to be revised for grammatical errors and syntax. After careful English proofreading the manuscript can be accepted for publication”

Answer:

Thank you very much for pointing this out.

To improve the linguistic quality of the text we involved a professional proofreader to carefully review and correct the text again. We hope the text will meet your requirements now.

Review Rohana Ghani (Reviewer 2):

Comment 1:

“Only some very minor typo errors to be corrected”

Answer:
Thank you for the gracious observations.

We involved a professional proofreader to carefully review and correct the text again to improve the quality of the text.